Author's response to reviews

Title: Population-based incidence trends of oropharyngeal and oral cavity cancers by sex among the poorest and underprivileged populations

Authors:

Ajit Auluck Dr (ajit_auluck@sfu.ca)
Blake Bryon Walker (bwalker@sfu.ca)
Greg Hislop (tgreghislop@gmail.com)
Scott Lear (SLear@providencehealth.bc.ca)
Nadine Schuurman (nadine@sfu.ca)
Miriam Rosin (rosin@sfu.ca)

Version: 2 Date: 10 January 2014

Author's response to reviews:

To,
The Editor
BMC Cancer

Dear Sir or Madam:

We have made the changes requested and please find below the details of the changes made.

1. Title page is inserted with full author details, their affiliations, email addresses, contact author details.

2. Abstract word length is 350 words and was structured as per the guidelines for authors.

3. Competing interests, financial disclosures, author's contributions and acknowledgements have been inserted in the title page.

Thank you very much for your feedback and help with this process.

Thanks
Ajit